Author’s response to reviews

Title: Rubinstein-Taybi syndrome in a Saudi boy with distinct features and variants in both the CREBBP and EP300 genes: A case report

Authors:

Mohammad Al-Qattan (moqattan@hotmail.com)
Abdulaziz Jarman (ajarman@kfshrc.edu.sa)
Atif Rafique (drafiq_2@yahoo.com)
Zuhair Al-Hassnan (zhassnan@kfshrc.edu.sa)
Heba Al-Qattan (heba.qattan96@gmail.com)

Version: 4 Date: 24 Dec 2018

Author’s response to reviews:

MGTC-D-18-00379R3

Rubinstein-Taybi syndrome in a Saudi boy with distinct features and variants in both the CREBBP and EP300 genes: A case report

BMC Medical Genetics

Dear Editor

We revised the manuscript as follows:

Editor Comments:

We do agree to extend the deadline of revision submission to allow authors response to the following concerns:

a) reviewer #3 (Jennifer Below) asked authors to provide the Sanger sequencing results of both variants.
please add them as figure, including the sequencing of parents as well.

Response: See new Figure 5

b) authors stated that all differences from the reference sequences (sequence variants) were assigned to one of the five interpretation categories as per the ACMG guidelines.

please categorize both variants according to five interpretation of ACMG guidelines, and explain briefly why authors suggest so.

Response: See text changes in lines 122-133